Author's response to reviews

Title: Clinical course, costs and predictive factors for response to treatment in carpal tunnel syndrome: The PALMS study protocol

Authors:

Christina Jerosch-Herold (c.jerosch-herold@uea.ac.uk)
Lee Shepstone (l.shepstone@uea.ac.uk)
Edward Wilson (ew442@medschl.cam.ac.uk)
Tony Dyer (tony.dyer@uea.ac.uk)
Julian Blake (julian.blake@nnuh.nhs.uk)

Version: 2
Date: 5 February 2014

Author's response to reviews:

Dear Editor

as requested in your email of 4th February 2014 I have now added an acknowledgements section, which includes the source of funding for the study for the first author. As this is at protocol stage there is no one else to be acknowledged for data acquisition or recruitment. I can also confirm that we did not use a language editor or scientific writer.

Yours sincerely

Christina Jerosch-Herold